

John Singh Cheema

The Mass Growth Factor – Snowball Effects in Aircraft Design

TYP DES DOKUMENTS | TYPE OF THE DOCUMENT Studienarbeit | Study Thesis

Nachnutzung | Reuse

Diese Publikation steht unter der Creative-Commons-Lizenz Namensnennung - Nichtkommerziell - Weitergabe unter gleichen Bedingungen 4.0 International (CC BY-NC-SA 4.0 International). Sofern der Name der Autor*innen/Rechteinhaber*innen genannt wird, kann der Inhalt vervielfältigt, verbreitet und öffentlich aufgeführt werden. Außerdem darf der Inhalt verändert werden (darunterfallen u. a. Übersetzungen des Werkes). Die Weitergabe der veränderten Fassung muss unter derselben Lizenz erfolgen. Der Inhalt darf nicht für kommerzielle Zwecke verwendet werden. Weitere Informationen und die vollständigen Bedingungen der Lizenz finden Sie hier: https://creativecommons.org/licenses/by-ncsa/4.0/legalcode.





SAIL HAW Hamburg Studentisches Publikationsportal Austausch – Information – Lernen



Project

The Mass Growth Factor – Snowball Effects in Aircraft Design

Author: John Singh Cheema

Supervisor:Prof. Dr.-Ing. Dieter Scholz, MSMESubmitted:31.03.2020

Faculty of Engineering and Computer Science Department of Automotive and Aeronautical Engineering DOI: https://doi.org/10.15488/11530

URN: https://nbn-resolving.org/urn:nbn:de:gbv:18302-aero2020-03-31.011 Associated URLs: https://nbn-resolving.org/html/urn:nbn:de:gbv:18302-aero2020-03-31.011

 $\ensuremath{\mathbb{C}}$ This work is protected by copyright

The work is licensed under a Creative Commons Attribution-NonCommercial-ShareAlike 4.0 International License: CC BY-NC-SA https://creativecommons.org/licenses/by-nc-sa/4.0

Any further request may be directed to: Prof. Dr.-Ing. Dieter Scholz, MSME E-Mail see: <u>http://www.ProfScholz.de</u>

This work is part of: Digital Library - Projects & Theses - Prof. Dr. Scholz http://library.ProfScholz.de

Published by Aircraft Design and Systems Group (AERO) Department of Automotive and Aeronautical Engineering Hamburg University of Applied Science

This report is deposited and archived:

- Deutsche Nationalbiliothek (<u>https://www.dnb.de</u>)
- Repository of Leibniz University Hannover (<u>https://www.repo.uni-hannover.de</u>)
- Internet Archive (<u>https://archive.org</u>) Item: <u>https://archive.org/details/TextCheemaProject.pdf</u>

This report has associated published data in Harvard Dataverse: <u>https://doi.org/10.7910/DVN/6NHDDP</u>

Abstract

Purpose – This project work shows a literature survey, clearly defines the mass growth factor, shows a mass growth iteration, and derives an equation for a direct calculation of the factor (without iteration). Definite values of the factor seem to be missing in literature. To change this, mass growth factors are being calculated for as many of the prominent passenger aircraft as to cover 90% of the passenger aircraft flying today. The dependence of the mass gain factor on requirements and technology is examined and the relation to Direct Operating Costs (DOC) is pointed out.

Methodology – Calculations start from first principles. Publically available data is used to calculate a list of mass growth factors for many passenger aircraft. Using equations and the resulting relationships, new knowledge and dependencies are gained.

Findings – The mass growth factor is larger for aircraft with larger operating empty mass ratio, smaller payload ratio, larger specific fuel consumption (SFC), and smaller glide ratio. The mass growth factor increases much with increasing range. The factor depends on an increase in the fixed mass, so this is the same for the payload and empty mass. The mass growth factor for subsonic passenger aircraft is on average 4.2, for narrow body aircraft 3.9 and for wide body aircraft (that tend to fly longer distance) 4.9. In contrast supersonic passenger aircraft show a factor of about 14.

Practical implications – The mass growth factor has been revisited in order to fully embrace the concept of mass growth and may lead to a better general understanding of aircraft design. **Social implications** – A detailed discussion of flight and aircraft costs as well as aircraft development requires detailed knowledge of the aircraft. By understanding the mass growth factor, consumers can have this discussion with industry at eye level.

Originality/value – The derivation of the equation for the direct calculation of the mass growth factor and the determination of the factor using the iteration method for current aircraft was not shown in the examined literature.



The Mass Growth Factor -Snowball Effects in Aircraft Design

Task for the project

Background

The mass growth factor is fundamental for the preliminary design of the aircraft. It is usually defined as the ratio of an increase in total mass (take-off mass) due to an arbitrary increase in local mass (empty mass), which is determined after a complete iteration in the aircraft design to achieve the original performance requirements (payload and range). The iteration of the aircraft construction provides a further increase of the take-off mass after each loop, so that an initial (local) mass increase worsens the situation like a snowball turning into an avalanche. Hence, the pseudonym snowball effect. The concept of the mass growth factor is probably as old as aviation. It was intensively discussed from the 1950s to the 1970s and is still mentioned repeatedly today. Nevertheless, it seems not to be understood well enough today. Perhaps its importance has decreased due to modern computing power, which provides quite accurate mass estimates in every design phase, but keeps the engineer from having a feel for the numbers.

Task

The task of this project work is to determine the mass growth factor. This shall be done by using the iteration method. Furthermore an equation shall be derived, with which the mass growth factor can be calculated directly. The calculation methods shall be used to calculate mass growth factors for many of the common passenger aircraft. The following sub items shall be considered:

- Literature research: Collecting previous findings
- Derive methods for calculating the mass growth factor
- Determining the mass growth factor for passenger aircraft flying today
- Sensitivity test of the mass growth factor

The results should be documented in a report. When preparing the report, the relevant DIN standards must be observed.

Content

	Page
Abstract	
List of Figures	7
List of Tables	
List of Symbols	9
List of Abbreviations	
Definitions	

1	Introduction	13
1.1	Motivation and Objectives	13
1.2	Title Terminology	14
1.3	Literature Review	14
1.4	Structure of the Work	15
2	Fundamentals and Literature Review	16
2.1	Risse 2016	16
2.2	Sinke 2019	18
2.3	Fürst 1999	19
2.4	SAWE 2019	20
2.5	Ballhaus 1954	20
2.6	Saelman 1973	23
2.7	Howe 2000	24
2.8	Jenkinson 1999	25
2.9	Müller 2003	25
2.10	Roskam 1989	26
2.11	Torenbeek 1988	27
3	Mass Growth Factor by Iteration	29
3.1	Equations	29
3.2	Example: Boeing 767-300	31
3.3	Mass Growth Factor for 90% of Current Aircraft	33
4	Mass Growth Factor Beyond Iteration	36
4.1	Generalization of Iteration Results	36
4.2	Mass Growth Factor from Operating Empty Mass and Fuel Mass Fraction	39
4.3	Mass Growth Factor from Payload Fraction	41
4.4	Mass Growth Factor Depending on Technology and Range Requirements	43
4.5	Mass Growth Factor Sensitivity with Range	45
4.6	Mass Growth Factor Sensitivity with Technology	47
4.7	Mass Growth Factor as an Economical Indicator	48
4.8	The Problem of a Large Mass Growth Factor	50

5 I	Discussion	2
6 8	ummary and Recommendations54	1
List of Refe	rences	5
Appendix A	Aircraft Making Up 90% of the World Fleet6	l
Appendix B	Mass Growth Factor for Four Aircraft Categories	2
Appendix C	Derivation: Mass Growth Factor – Inverse of Payload Fraction	3

List of Figures

Figure 2.1	Influence of the variation of the MTOW on the compliance with the	
C	TLARs (according to Risse 2016, Chapter 3, p.55).	17
Figure 2.2	Representation of the mass snowball effect (according to	
C	Sinke 2019,p.54)	18
Figure 2.3	Influence of design assumptions on mass growth (according to	
C	Howe 2000, Chapter 9, p.280)	24
Figure 4.1	Mass growth factor of old long-range aircraft (Screenshot:	
C	Evaluation_90%_aircraft_+_category.xlsm)	38
Figure 4.2	Mass growth factor of new short-range aircraft (Screenshot:	
-	Evaluation_90%_aircraft_+_category.xlsm)	38
Figure 4.3	Operating empty mass fraction as a function of design range (according	
-	to Scholz 2015, Section 5, p.29)	39
Figure 4.4	Payload fraction as a function of maximum take-off mass (according to	
	Scholz 2018, slide 30)	40
Figure 4.5	Mass growth factor and mass fraction over the range (Screenshot:	
	Sensitivity_with_Range.xlsm)	47

List of Tables

Table 2.1	Influence of aircraft size on mass parameters (according to Jenkinson	25
Table 2.2	Effect of a 10% increase in structural weight on the maximum take-off weight at constant mission (according to Torenbeek 1988,	. 25
TUL 31	Chapter 8, p.266)	.27
I able 3.1	Masses of the Boeing /6/-300 (according to Jenkinson 2019)	. 31
Table 3.2	Further iteration steps and results	. 32
Table 3.3	Evaluation of 90% of all current flying commercial aircraft including	
	two supersonic aircraft	. 34
Table 4.1	Mass growth factor for different aircraft categories	. 37
Table 4.2	Mass growth factor for typical mass fractions	.41
Table 4.3	Sensitivity Test - Parameters for the A320-200	. 45
Table 4.4	Results of sensitivity test depending on the requirement	. 45
Table 4.5	Results of sensitivity test depending on the technology	. 48
Table 4.6	Results of sensitivity test depending on technology - with reduction	. 48
Table A.1	90% of current aircraft (according to Robson 2019)	. 61
Table B.1	Categorization of current aircraft (Screenshot:	
	Evaluation_90%_aircraft_+_category.xlsm)	. 62

List of Symbols

$m_{take-off}$	Take-off mass
m _{drive}	Mass of the drive system
$m_{structure}$	Mass of the structure
m _{fuel}	Fuel mass
W_p	Propulsion system weight
W_{g}	Gross weight
W_{f}	Fuel weight
W _{str}	Structure weight
W_F	Fuel weight
W_{S+VE}	Weight of Structure and Variable Equipment
W	Gross weight
F	Fixed weight
S	Size
n	Strength or structural criteria
Q	Quality
P MZ	Cross weight
vv _g	GIOSS Weight
W_0 m_	Fixed weight
	Maximum Payload
$m_{\rm PI}$	Pavload
m_{MTO}	Maximum take-off mass
m_{OE}	Operating empty mass
Δm_L	Local mass growth
Δm_G	Global mass growth
k _{MGF}	Mass growth factor
m_{ZP}	Zero Passenger Weight
С	Specific fuel consumption
R	Range
В	Breguet factor
Ε	Glide ratio in cruise flight
v _{cr}	Cruising speed
m_1	Take-off mass
m_2	Landing mass
C_{DEP}	Depreciation
C_{INT}	Interest
C _{INS}	Insurance
C_F	Fuel
C_M	Maintenance
C_{C}	Crew

C_{FEE}	Fees and charges
$C_{a/c,a}$	Aircraft annual costs
n _{t,a}	Number of Flights per year
$C_{s,m}$	Seat-mile costs
$C_{a/c,t}$	Aircraft trip cost
n_{PAX}	Number of passengers
U _{a,f}	Annual aircraft use
t _f	Defined aircraft time

Indices

MTO	Maximum Take-Off
OE	Operating empty
PL	Payload
cr	Cruise
a/c	Aircraft
a	Annual Costs
t	Flight, cycle or trip

List of Abbreviations

SAWE	Society of Allied Weight Engineers		
LTH	Luftfahrttechnisches Handbuch		
MGW	Mass growth		
TLAR	Top Level Aircraft Requirements		
DOC	Direct Operating Costs		
GF	Growth Factor		
MD	McDonnell Douglas		
А	Airbus		
ATR	Avions de Transport Régional		
DHC	De Havilland Canada		
neo	New Engine Option		
PAX	Passenger		
SFC	Specific Fuel Consumption		
TU	Tupolev		

Definitions

Preliminary sizing

The project phase of aircraft development consists of the activities preliminary sizing and design. The most important design parameters are determined in the preliminary sizing of the aircraft.

The preliminary sizing according to ... can be carried out without the need for an exact aircraft geometry. However, there should already be some ideas about the configuration to be chosen and the propulsion system. Only then reasonable assumptions about the parameters to be selected can be made. (Scholz 2015, Section 2, p.1)

Maximum take-off mass

The maximum take-off mass is the maximum mass of an aircraft that may be used for a take-off. It is the operating empty mass plus the maximum permissible combination of payload and operating materials (fuel). (DIN 9020, Scholz 2015, Section 10)

Operating empty weight

The operating empty weight is the take-off weight of an aircraft minus the payload and the operating materials. In other words, the operating empty weight is the manufacturer's empty weight (structure, propulsion system, standard equipment) plus mass deviations, fixed and mobile operating equipment (DIN 9020, Scholz 2015, Section 10)

Fuel

Aviation fuels are specialized types of petroleum-based fuel used to power aircraft. They are refined to a higher specification under greater quality control than fuels used in less critical applications, such as heating or road transport and often contain additives to reduce the risk of liquid fuel icing at low temperatures or the explosion of fuel vapor at high temperatures. (SKYbrary 2020)

Payload

The payload of an aircraft or spacecraft is the amount or weight of things or people that it is carrying. (Collins 2020)

Sensitivity

Synonyms for sensitivity: sensitiveness, delicacy... (Duden 2019)

1 Introduction

1.1 Motivation and Objectives

Hardly any other market segment is developing as dynamically as industrial lightweight construction. Even in today's aircraft development, it is this construction method that is most important. Less weight means lower fuel consumption. Lightweight aircraft are therefore more economical and environmentally friendly. The most important requirement in aircraft construction, in addition to the relevant construction regulations for the certification of aircraft, is weight reduction.

In the project phase of aircraft development, an aircraft is dimensioned taking into account the requirements and boundary conditions. This means that during this phase one tries to fulfill a certain performance catalog. Design changes can lead to the growth of mass. If an increase of the operating empty mass is necessary, this will result in a revision of the design draft. This in turn influences the take-off mass.

This is where this project work starts and deals with the topic of mass increase and the resulting mass snowball effect. An increase of the operating empty mass causes a further increase in this area. This leads to an iterative process. This also applies to the reduction of the operating empty mass, which leads to mass reductions. The growth factor can be used to determine the extent to which an increase in the operating empty mass affects the take-off mass of the aircraft.

The development of modern design synthesis programs has made it possible to quickly determine the influence of weight changes on the overall configuration. Although this has somewhat reduced the need to use mass growth factors to determine weight change effects, they are still widely used in rapid conceptual trade studies and as plausibility checks of analysis results.

The aim of this project is to find out how the mass growth factor can be determined. This shall be done by applying the "first law of aircraft design", which is used to estimate the maximum take-off mass. The goal is to show a mass growth iteration and to derive an equation for a direct calculation of the factor.

In this context, the calculation methods serve to calculate the mass growth factors for many of the common passenger aircraft. In addition, previous findings on this topic are to be cited and new ones gained. Among other things, the dependence of the mass growth factor on requirements and technology will be investigated.

1.2 Title Terminology

The title of the project work is "The Mass Growth Factor - Snowball Effects in Aircraft Design". Each of the terms contained in the title is defined below.

Mass growth factor

The so-called mass growth factor quantifies, how much the take-off mass is finally increased, provoked by a unit mass increase (e.g., 1kg) of the empty mass due to the original design change (without iteration). Its value strongly varies with aircraft type and requirements... (Risse 2016, S.54f.)

Snowball effect

In aerospace engineering, it [the snowball effect] is used to describe the multiplication effect in an original weight saving. A reduction in the weight of the fuselage will require less lift, meaning the wings can be smaller. Hence less thrust is required and therefore smaller engines, resulting in a greater weight saving than the original reduction. This iteration can be repeated several times, although the decrease in weight gives diminishing returns. (Wiki 2018)

Aircraft Design

The aircraft design process is generally divided into three phases: Project phase, definition phase and development phase.

According to Scholz 2015 (Section 1, p.1) aircraft design is described as follows:

The task of aircraft design in the practical sense is to 'provide the geometric description of a new aircraft'. The new aircraft is described by a three-view drawing, a fuselage crosssection, a cabin layout, and a list of aircraft parameters.

1.3 Literature Review

Risse (2016) provides as an introduction to the topic. In this project, a short presentation of the topic is given, including the aeronautical manual from Fürst (1999), which in turn refers to a SAWE manual. SAWE (Society of Allied Weight Engineers) is a society of engineers specialized in the field of mass properties.

In Section 2 of this project, a literature search is conducted first. For this purpose, the book by SAWE (2019) as well as papers (Ballhaus 1954 and Saelman 1973) on the subject of the mass growth factor are consulted. Furthermore, books from the field of aircraft design are used to provide further insights on this topic. This literature includes Jenkinson 1999, Howe 2000, Müller 2003, Roskam 1989 and Torenbeek 1988.

The reference to the lecture notes according to Scholz 2015 occurs repeatedly at various points throughout the work. Most of the information from this script is taken from Section 5, which deals with preliminary sizing.

The research of the necessary data about the masses to determine the mass growth factor for the different aircraft comes from the books Taylor 1989, Lambert 1991 and Jackson 2011. Detailed and reliable data can be obtained from these books. For data not available in these books, further different sources are used, which are listed in the corresponding chapter. These include Jenkinson 2019 and Airbus and Boeing manufacturer websites.

1.4 Structure of the Work

This work is divided into the following sections:

- Section 2 describes the basics of the mass snowball effect and mass growth factor. Previous investigations or findings on this topic are cited by means of literature research.
- Section 3 deals with the determination of the mass growth factor using the "first law of aircraft design" and the iteration method. With this calculation method, the factor is calculated for so many known passenger planes that they cover 90% of the passenger planes flying today.
- Section 4 first presents the results derived from the previous section. For typical mass fractions, values for the mass growth factor are given. An equation for a direct calculation of the factor is derived. Furthermore, the dependence of the mass growth factor on requirements and technology is examined and the relationship to direct operating costs (DOC) is highlighted.
- Section 5 discusses the main results of this work
- Section 6 summarizes the most important results of this work and gives an outlook.
- Appendix A includes the evaluation of current passenger aircraft.
- Appendix B contains the results of the mass growth factor for aircraft categories.
- **Appendix C** contains a part of the derivation of the equation for a direct calculation of the factor.

2 Fundamentals and Literature Review

The determination and investigation of the mass growth factor is carried out using known formulas from the preliminary sizing of aircraft from the field of aircraft design. The basic knowledge for this shall not be repeated in this project and is a prerequisite. If necessary, the lecture notes Scholz 2015 are recommended.

2.1 Risse 2016

The design of an aircraft is an iterative process. A first assumption, which describes the maximum take-off mass, is necessary to get a first preliminary sizing. Then the operating empty mass can be estimated. Based on this operating empty mass and the design payload it is possible to estimate the required fuel. A new assumption, which describes the maximum take-off mass, can be generated by summing the operating empty mass, payload and fuel mass. This design loop has to be executed until convergence is reached. Requirements and boundary conditions have to be met.

For aircraft that are still in the design process or already in operation, the mass can increase due to specific changes in design or structure and the integration of new technologies or systems. If the assumptions in the design process are too optimistic, the planned values may be exceeded and finally the weight of the aircraft may be too high. The operating empty mass of an aircraft in service increases during its life cycle. This occurs when basic equipment (new technologies) or operating equipment (for example, a new entertainment system) is retrofitted. The operating empty mass also increases due to repairs to the aircraft.

As the operating empty mass of the aircraft increases, the fuel consumption in a given design area increases. The increase in fuel mass may even require larger fuel tanks, which primarily increases the maximum take-off mass. For structural components (e.g. wings), this can in turn result in further adjustments by increasing flight or ground loads. The thus increased component masses again result in an increase of the fuel mass and consequently an increase of the take-off mass, which results in an iterative process (according to Risse 2016, Chapter 3, p.54f.).

The iterative interaction between the masses of the components of the structure, system or drive and the take-off mass is generally referred to as the mass snowball effect (according to Risse 2016, Chapter 3, p.54f.).

The so-called mass growth factor quantifies how much the take-off mass is finally increased, caused by an increase in the operating empty mass due to the original design change. Its value varies greatly depending on the aircraft type and requirements (according to Risse 2016, Chapter 3, p.54f.).



Figure 2.1 Influence of the variation of the MTOW on the compliance with the TLARs (according to Risse 2016, Chapter 3, p.55)

Figure 2.1 shows the consequence of a significant variation of the take-off mass in the form of a loss of performance or even a difference to the requirements. If the geometry of the airframe and the engines remains unchanged, an increase of the maximum take-off mass (MTOW) results in an increased wing loading (W/S) and a reduced thrust-to-weight ratio (T/W) (according to Risse 2016, Chapter 3, p.55f.).

The curves of the limitation, which show the requirements for take-off, climb, cruise flight and landing, contain a valid size range. Within this range an optimal combination of T/W and W/S must be determined. Deviation from this (ideal) design point usually results in reduced overall performance and may cause requirements (TLARs) not to be met (according to Risse 2016, Chapter 3, p.55f.).

In Figure 2.1, the selected sizing point is located near the boundary lines. There is a susceptibility to deviations for the start as well as for the climb requirements with increasing mass (increasing MTOW). This can result in necessary design adjustments (according to Risse 2016, Chapter 3, p.55f.).

The reverse case of a later mass reduction becomes particularly relevant for a retrofit design, e.g. for the integration of a fuel-saving technology. If the airframe and engine components are kept constant again, the retrofitted design with reduced MTOW has a reduced W/S and an increased T/W. This is rather uncritical with regard to compliance with the TLARs. However, a lower wing loading can also result in a loss of performance, and an increased T/W can include a drive system that is too heavy (according to Risse 2016, Chapter 3, p.55f.).

2.2 Sinke 2019

As described in the previous section, the increase in operating empty mass brings with it further increases. This leads to a so-called mass snowball effect, which in turn leads to a build-up of snow.

In Figure 2.2. the mass snow ball effect is displayed. In this view it is assumed that the fuel mass and payload do not change.



Figure 2.2 Representation of the mass snowball effect (according to Sinke 2019, p.54)

If the weight of e.g. structures or systems is increased, this leads to an increase of the total aircraft weight. This leads to

- \rightarrow greater required lift
- \rightarrow larger wings
- \rightarrow this increases the resistance, including aerodynamic resistance
- \rightarrow therefore the thrust must be increased
- \rightarrow this leads to larger engines
- \rightarrow this increases the weight again (according to Sinke 2019, p.54)

This process is iterative and continues until the weight converges. If the weight is reduced at the beginning, this leads to further reductions of the sizes shown in Figure 2.2 above.

2.3 Fürst 1999

The manual first describes the mass growth factor. According to this, the mass growth factor states what effect "an increase in empty mass of e.g. 1.0 kg has on the total mass of the aircraft" (Fürst 1999, p.1). The smallest possible value for the mass growth factor is 1.0. "However, this value is only achieved if a reduction in performance caused by the increase in mass is accepted." (Fürst 1999, p.1). As a rule, the mass growth factor is greater than 1.0, since the mass increase will be higher at the take-off mass than at the empty mass. If an increase of the empty mass is necessary in the conception and preliminary design phase of an aircraft, this has an influence on the take-off mass. In addition, the design draft must be revised.

Furthermore, the description of the mass snowball effect is given. Under the assumption that the range of the aircraft remains the same, the correlations with the mass growth factor are to be presented clearly.

"If the range is to remain constant, each increase in mass requires an additional amount of fuel. Increased fuel volume and associated enlarged fuel tanks are also necessary. Structural components that are dimensioned according to the flight loads become heavier the more the take-off mass increases. Changes in volume and structure in turn result in a larger quantity of fuel, which in turn increases the take-off mass. This process is iterative and continues until the interdependent functions - increase of the take-off mass and the resulting increase in mass of structure and fuel - converge." (Fürst 1999, p.1).

The empty mass of an aircraft can be divided into variable and fixed mass components. The variable components, such as the structure, are components "whose mass is influenced by the respective take-off mass" (Fürst 1999, p.1). Fixed mass components, such as avionics, include components "whose mass is not affected by the take-off mass" (Fürst 1999, p.1). Both mass increases in the variable and in the fixed mass component of the empty mass influence the level of the mass growth factor.

The mass growth factor MGF can be determined with the following equation:

$$MGF = \frac{1}{1 - \left(\frac{m_{drive}}{m_{take-off}} + \frac{m_{structure}}{m_{take-off}} + \frac{m_{fuel}}{m_{take-off}}\right)}$$
(2.1)

Finally, values for the mass growth factor are given for certain types of aircraft. A subsonic transport aircraft with a moderate set of requirements has such a factor of about 1.5, whereas for a fighter aircraft with a demanding set of requirements this factor is much higher.

"For most aircraft, the factors are between 3.0 and 5.0. The average value for subsonic aircraft is 3.4 and for supersonic aircraft 4.6. For already built vertical take-off aircraft or some STOVL projects, mass growth factors of 2.5-5.0 have been determined." (Fürst 1999, p.1).

2.4 SAWE 2019

According to the definition in this book, the mass growth factor is the increase in total weight required per pound of empty weight. Here, the mass snowball effect is also described as an iterative process. As the load increases, the addition to the total weight results in an increase in structural components. In addition, additional volume and an adequate fuel absorption structure are necessary. The integration of volume and structure increases the fuel requirement, which leads to an even higher total weight. This process is continued until there is a convergence between the increase in weight of the components and increased fuel and total weight.

This is followed by a description of the classic mass growth factor definition. According to this definition, the mass growth factor determines how the total weight increases for each additional pound of the fixed weight portion of the empty weight. Two categories of empty weights can be identified. One changes when the total weight is changed (variable weight components), the other is not affected by changes in total weight (fixed weight components). The fixed weight category includes items such as avionics. The variable weight category includes structural elements.

The mass growth factor GF (growth factor) is calculated as follows:

$$GF = \frac{1}{1 - \left(\frac{W_p}{W_g} + \frac{W_{str}}{W_g} + \frac{W_f}{W_g}\right)}$$
(2.2)

In the classical mass growth factor definition, the total weight growth, caused by changed components with fixed weight, is taken into account. A similar effect is also produced by modified components with variable weight. For example, an increase in wing area will result in an increase in the total weight of the configuration, which is necessary to maintain performance.

2.5 Ballhaus 1954

In this paper, the mass growth factor is rejected as the ratio of the total weight to the given fixed weight.

$$G = \frac{\text{Change in Gross Weight}}{\text{Fixed Weight Added or Change in Fixed Weight}}$$
(2.3)

First, the relationship between a change in weight and the requirements of the aircraft is explained. An aircraft is designed taking into account certain design requirements and specifications. The aircraft will therefore have a specific engine, total weight and specific geometric, structural and aerodynamic layout. If, for any reason, the fixed weight changes, the total weight of the aircraft must be adjusted so that the aircraft still meets the specified requirements exactly.

For a more precise determination of the mass growth factor, the mathematical derivation is used. In this method, the total weight is divided into fixed weight, which does not change with total weight, size, thickness and quality, and variable weight, which changes with the mentioned quantities.

The following therefore applies to the gross weight:

$$Gross weight = Fixed weight + Variable weight$$
(2.4)

$$W = F + f(W, S, n, Q, P)$$
 (2.5)

Here is f(W, S, n, Q, P) the function that combines the variable weight with W, S, n, Q and P in relation to each other.

Reordering of Equation (2.5) leads to:

$$F = W - f(W, S, n, Q, P)$$
 (2.6)

And by differentiation follows:

$$\frac{dF}{dW} = 1 - \frac{df}{dW} \tag{2.7}$$

With the connection (2.3) the mass growth factor is obtained:

$$G = \frac{dW}{dF} \tag{2.8}$$

Respectively

$$G = \frac{1}{\frac{dF}{dW}}$$
(2.9)

By inserting Equation (2.7) into (2.9) finally follows:

$$G = \frac{1}{1 - \frac{df}{dW}}$$
(2.10)

This equation is a general expression for the mass growth factor. The factor then depends only on the variable weight in the aircraft and especially on the total derivative of the variable weight in relation to the total weight.

However, since the mass growth factor only exists with constant strength criteria (constant quality and performance), the variable weight only depends on the total weight and size. The new expression is g(W, S) and is the function that combines the variable weight W and S in relation to each other.

The variable weight is made up of the weight of the propulsion system, the fuel weight and the weight of the structure and variable equipment. The new expression for the variable weight is then:

$$g(W,S) = W_P + W_F + W_{S+VE}$$
(2.11)

The power, the thrust-to-weight ratio, the wing loading and the thrust-specific weight of the drive system are assumed to be constant. W_P and W_F depend only on the total weight, whereas W_{S+VE} depends on both, the total weight and the size. The new total derivative of the variable weight in relation to the total weight is

$$\frac{dg}{dW} = \frac{\partial W_P}{\partial W} + \frac{\partial W_F}{\partial W} + \frac{dS}{dW} \frac{\partial W_{S+VE}}{\partial S} + \frac{\partial W_{S+VE}}{\partial W}$$
(2.12)

The mass growth factor for constant strength criteria (constant quality and performance) is then obtained:

$$G = \frac{dW}{dF} = \frac{1}{1 - \left[\frac{\partial W_P}{\partial W} + \frac{\partial W_F}{\partial W} + \frac{S}{W}\frac{\partial W_{S+VE}}{\partial S} + \frac{\partial W_{S+VE}}{\partial W}\right]}$$
(2.13)

In order to determine the mass growth factor, mass fractions are used in the total derivation of variable weight. This is illustrated by the following example, in which the mass growth factor is calculated for an aircraft with a range of 5000 NM. The aircraft has the following masses:

W = 100000 lbs $W_F = 50000 \text{ lbs}$ $W_P = 15000 \text{ lbs}$

This results in the following mass fractions of $\frac{W_F}{W} = 0.50$ and $\frac{W_P}{W} = 0.15$.

The remaining sizes are given as $\frac{s}{w} \frac{\partial W_{S+VE}}{\partial s} = 0.14$ and $\frac{\partial W_{S+VE}}{\partial W} = 0.09$.

For the aircraft, Equation (2.13) gives a mass growth factor of:

$$G = \frac{1}{1 - [0.15 + 0.50 + 0.14 + 0.09]} = 8.33$$

Next, the effect of increasing the range is mentioned. If the range of the aircraft is to be increased from 5,000 NM to 6,000 NM, more fuel must be carried in addition. For this purpose, 6,500 lbs are subtracted from the fixed weight and 6,500 lbs more fuel is added. The fuel mass is then $W_F = 56,500$ lbs and the fuel mass ratio accordingly $\frac{W_F}{W} = 0.565$. The new mass growth factor is then obtained:

$$G = \frac{1}{1 - [0.15 + 0.556 + 0.14 + 0.09]} = 18.18$$

With an increase in range and a corresponding increase in fuel mass, the mass growth factor increases.

2.6 Saelman 1973

According to the definition of this paper, the mass growth factor is the ratio in which the total weight is increased to any increase in weight in the aircraft.

At a certain point the mass growth factor β is given via differentiation as follows:

$$\beta = \frac{dW_g}{dW_o} = \frac{dgross \ weight}{dfixed \ weight}$$
(2.14)

The fixed weight includes the payload, the weight of crew and passenger accommodation, the pressure critical cabin structure, the flight station, smaller fuselage frames etc.

To develop a mathematical term for the mass growth factor, the variation of the weights of the central aircraft components with the growth weight is determined. Finally, the total weight can be calculated as the sum of the fixed weight and combined terms that are in proportion to the take-off weight performance.

A resulting formula describes the total weight as the sum of fixed weight and combined factors. Finally, an expression for the mass growth factor is derived by implicitly differentiating the total weight according to the fixed weight.

As a result, the paper shows that the mass growth factor increases when the growth weight is increased. It should be noted that a relatively small fraction of the payload or fixed weight in the total weight results in a high mass growth factor. Wide-body commercial aircraft have

growth factors of 3, whereas the SST has growth factors in the range of 6 to 7, combat aircraft have growth factors in the range of 10, and rockets and spacecraft have exceptionally high growth factors.

2.7 Howe 2000

This book shows the connection between design assumptions and mass growth. Accordingly, the analysis of design work is an iterative process. If there is sufficient justification for the initial assumptions, the process will converge to a final design with features that can be identified with sufficient accuracy. In an analysis at a later point in time it may turn out that some assumptions were not justified. A deviation of the design is possible in this case.

Figure 2.3 is a simple example of how this can be done.





Influence of design assumptions on mass growth (according to Howe 2000, Chapter 9, p.280)

The black line represents the original design configuration. A certain design mass is assumed. If the assumptions differ, e.g. due to an increase in weight, the dashed line is generated. Then a mass increase (modified mass) occurs. However, if the mass is to correspond to the originally assumed design mass, a reduction in performance must be expected or the requirements must be reduced.

In case of deviations from the requirements, it is necessary to rethink the overall concept, possibly to investigate alternative overall configurations or to review the dominant performance requirements.

2.8 Jenkinson 1999

This book starts with an explanation of the mass snowball effect. If the aircraft specifications (e.g. take-off performance, range, etc.) are met, the structural weight increase will result in higher fuel consumption, larger engines, larger wing and empennage surfaces and stronger landing gear due to design inefficiency.

A mass growth factor of 3 is cited, which means that per kg of additional structural mass on the aircraft, the maximum take-off mass of the aircraft increases by approximately three kg. The influence of the weight increase is illustrated using two aircraft. The range of both aircraft is identical, but the number of passengers during flight is different. Table 2.1 shows the passenger numbers and the mass parameter changes of the two aircraft.

Table 2.1	Influence of aircraft size on mass parameters (according to Jenkinson 1999, Chapter 7,
	p.128)

300	600
1000	1000
1879	1756
1255	1431
3034	3188
3.03	3.19
0.87	1.19
	300 1000 1879 1255 3034 3.03 0.87

The structural mass was increased by 1000 kg for both aircraft and the aircraft were redesigned. The aircraft with 300 seats has a mass growth factor of 3.03 and the aircraft with 600 seats has a factor of 3.19. The larger aircraft therefore has a larger mass growth factor with the same increase in structural mass.

2.9 Müller 2003

In this book the problems of mass calculation and the sensitivity of mass calculation for the design are explained. The problem lies in the fact that the different weight formulas must be derived from actual masses, which in turn depend on the quality of construction and building. The sensitivity is based on the fact that mass overruns always result in losses in performance.

The problem is that the planned values are exceeded if the assumptions are too optimistic and that the aircraft becomes too large and heavy if the assumptions are too skeptical. If the wings are too large and the cell components associated with them, the slightly higher frictional resistance results in a slightly lower maximum speed.

If flight performances are kept constant, additional weights and/or higher resistances result in a growth factor in the take-off mass. It is explained that one kg of additional weight in an aircraft results in a two kg increase in take-off mass. It is also stated that with 1 daN of additional drag, the take-off mass increases by about 3.5 kg.

2.10 Roskam 1989

The book first shows the existence of a linear relation between the decadic logarithm of the operating empty mass and that of the take-off mass. The operating empty mass of different types of aircraft was plotted against the take-off mass in diagrams, thus proving the correlation. By means of a regression analysis, trend lines were then formed in the diagrams.

It becomes obvious that a logarithmic function with the parameters A and B must exist, by which said trend lines are represented in the logarithmic diagrams. By this logarithmic function the operating empty mass can be determined by the take-off mass

The mass growth factor is then discussed. For this purpose, the change in take-off mass, depending on several parameters such as payload and operating empty mass, is analyzed. First of all, the mass growth factor is specified by payload. For its determination a formula of the ratio of take-off mass by payload is used, which itself depends on different parameters. This ratio derived results in the mass growth factor by payload.

The mass growth factor by operating empty mass is obtained by using the logarithmic function with the parameters A and B. This represents the connection between the take-off mass and the operating empty mass. If the ratio between the take-off mass and the operating empty mass is derived, the mass growth factor by operating empty mass is obtained. In an example with a jet airliner that has a take-off mass of approximately 126100 lbs, this is 1.93. This means that for each additional pound of operating empty mass, the takeoff mass must be increased by 1.93 lbs to ensure consistent mission performance.

2.11 Torenbeek 1988

First of all, an explanation is given of the importance of reducing the weight of an aircraft design, mainly in large aircraft of high complexity. The increase in weight of a component is accompanied by an additional weight elsewhere. This is described as the "snowball effect of weight growth". The possibilities of weight minimization depend on the phase of the design process.

In the conceptual design at the beginning, the selected aircraft layout, the chosen geometry and the choice of detailed configuration influence the weight. The evaluation of reductions or increases in weight is usually done for a constant design performance, except when the limited engine power does not allow it. Consequently, each increase in component weight is accompanied by a higher take-off weight. However, if the increased weight of the components is caused by a design change that contributes to the performance increase (e.g. improved high-lift systems), a reduced take-off weight may result.

Table 2.2 shows the sensitivities for increases in structural weight based on some typical missions. The effects of a structure weight increase of 10% on the maximum take-off weight for a constant mission can be seen.

AIRPLANE CATEGORY	DESIGN	RANGE	EFFECT ON MTOW
subsonic transport	250	nm	6.5 Z
	1000	nm	6.9 %
	3000	nm	7.0 %
supersonic transport	3000	am	9.4 %
transport VTOL	250	it M	6.9 %
military VTOL	250	nm	9.5 %

Table 2.2Effect of a 10% increase in structural weight on the maximum take-off weight at constant
mission (according to Torenbeek 1988, Chapter 8, p.266)

For a subsonic airliner with a design range of e.g. 250 NM, a 10% increase in weight of the structure leads to a 6.5% increase of the maximum take-off mass. It should also be noted that the sensitivity also depends on where the weight increase occurs.

With the help of this data from the book, a mass growth factor should now be determined. The 10% increase of the operating empty mass can be represented as:

$$\Delta m_{OE} = m_{OE} \cdot 0.1 \tag{2.15}$$

And the increase of the maximum take-off mass as

$$\Delta m_{MTO} = m_{MTO} \cdot x \tag{2.16}$$

To determine a factor, the ratio of the two mass increases is used:

$$\frac{\Delta m_{MTO}}{\Delta m_{OE}} \tag{2.17}$$

Inserting (2.15) and (2.16) into (2.17) leads to:

$$\frac{\Delta m_{MTO}}{\Delta m_{OE}} = \frac{m_{MTO} \cdot x}{m_{OE} \cdot 0.1} = \frac{\frac{x}{0.1}}{\frac{m_{OE}}{m_{MTO}}}$$
(2.18)

Now for x e.g. the 6.5% increase of the maximum take-off mass can be used. A value of 0.5 is assumed for the ratio of operating empty mass to maximum take-off mass. This results in a factor of:

$$\frac{\Delta m_{MTO}}{\Delta m_{OE}} = \frac{\frac{0.065}{0.1}}{0.5} = 1.3$$

Based on the data from Torenbeek 1988 and assuming that the operating empty mass fraction is 0.5, a mass growth factor of 1.3 is calculated.

3 Mass Growth Factor by Iteration

3.1 Equations

Previous studies indicate that when a local mass growth occurs, the increase in the take-off mass converges. The process of the mass snowball effect is iterative and continues until the take-off mass approaches a certain value. On the basis of this knowledge, a method for determining the mass growth factor is presented.

The maximum take-off mass m_{MTO} according to Scholz 2015 is composed of maximum payload m_{MPL} , fuel mass m_F (for the range at maximum payload) and the operating empty mass m_{OE} :

$$m_{MTO} = m_{MPL} + m_F + m_{OE} \tag{3.1}$$

The transformation of this formula results in the so-called "first law of aircraft design", which represents a relationship between the maximum take-off mass and the operating empty mass (according to Scholz 2015):

$$m_{MTO} = \frac{m_{MPL}}{1 - \frac{m_F}{m_{MTO}} - \frac{m_{OE}}{m_{MTO}}}$$
(3.2)

This equation is supplemented by the parameter Δm_L . This parameter contains the local mass growth. A local mass growth has the consequence that the total take-off mass growths. This global mass growth is expressed with the parameter Δm_G . This results in:

$$m_{MTO} + \Delta m_{G} = \frac{m_{MPL}}{1 - \frac{m_{F}}{m_{MTO}} - \frac{m_{OE} + \Delta m_{L}}{m_{MTO}}}$$
(3.3)

The mass growth factor k_{MGW} finally results from the ratio of global to local mass growth:

$$k_{MGW} = \frac{\Delta m_G}{\Delta m_L} \tag{3.4}$$

The local mass growth Δm_L depends on the weight by which the operating empty mass is increased. This can therefore be assumed to be known. In our case the local mass growth is $\Delta m_L = 1$ kg.

The determination of the global mass growth Δm_G is done by iteration. First, the Equation (3.1) is supplemented by the local and global growth of mass and named $m_{MTO.0}$:

$$m_{MTO,0} = m_{MTO} + \Delta m_G = m_{MPL} + m_{OE} + \Delta m_L + m_F$$
 (3.5)

This equation corresponds to Equation (3.3). For the first iteration step, the operating empty mass and the fuel mass are converted as mass fractions. It shall apply:

$$m_{OE} = \frac{m_{OE}}{m_{MTO}} \cdot m_{MTO} \tag{3.6}$$

And

$$m_F = \frac{m_F}{m_{MTO}} \cdot m_{MTO} \tag{3.7}$$

With the maximum take-off mass $m_{MTO,0}$ and the mass fractions, the first iteration step is done:

$$m_{MTO,1} = m_{MPL} + \frac{m_{OE}}{m_{MTO}} \cdot m_{MTO,0} + \Delta m_L + \frac{m_F}{m_{MTO}} \cdot m_{MTO,0}$$
(3.8)

The next iteration steps look like this:

$$m_{MTO,2} = m_{MPL} + \frac{m_{OE}}{m_{MTO}} \cdot m_{MTO,1} + \Delta m_L + \frac{m_F}{m_{MTO}} \cdot m_{MTO,1}$$
(3.9)

$$m_{MTO,3} = m_{MPL} + \frac{m_{OE}}{m_{MTO}} \cdot m_{MTO,2} + \Delta m_L + \frac{m_F}{m_{MTO}} \cdot m_{MTO,2}$$
 (3.10)

$$m_{MTO,4} = m_{MPL} + \frac{m_{OE}}{m_{MTO}} \cdot m_{MTO,3} + \Delta m_L + \frac{m_F}{m_{MTO}} \cdot m_{MTO,3}$$
(3.11)

The iteration is done until the maximum take-off mass converges to a value. The limit of the maximum take-off mass by increasing the operating empty mass by 1 kg is then $m_{MTO,X}$.

For the global mass growth Δm_G applies:

$$\Delta m_G = m_{MTO,X} - m_{MTO} \tag{3.12}$$

With this global mass growth Δm_G , the local mass growth, in our case of $\Delta m_L = 1$ kg, and using the formula (3.4), the mass growth factor k_{MGW} can finally be determined.

3.2 Example: Boeing 767-300

The following is an example for the calculation of the mass growth factor k_{MGW} with the abovementioned method for the Boeing 767-300. The local mass growth Δm_L is assumed to be 1 kg.

Using Jenkinson 2019, the following masses in Table 3.1 were determined for the aircraft.

Table 3.1Masses of the Boeing 767-300 (according to Jenkinson 2019)

Aircraft	m_{MTO} [kg]	m_{OE} [kg]	<i>m_{MPL}</i> [kg]
Boeing 767-300	156489	87135	39140

The fuel mass is also calculated with:

$$m_F = m_{MTO} - m_{MPL} - m_{OE} \tag{3.13}$$

The result is $m_F = 30214$ kg.

From these masses, the mass fractions can then be determined. These result in $\frac{m_{OE}}{m_{MTO}} = 0.5568123$ and $\frac{m_F}{m_{MTO}} = 0.19307427$.

First the maximum take-off mass $m_{MTO.0}$ is determined according to Equation (3.5). Inserting the values results in:

$$m_{MTO,0} = 39140 \, kg + 87135 \, kg + 1 \, kg + 30214 \, kg = 156490 \, kg$$

The first iteration step according to Equation (3.8) is given by inserting:

$$m_{MTO,1} = 39140 \ kg + 0.5568123 \cdot 156490 \ kg + 1 \ kg + 0.19307427 \cdot 156490 \ kg$$

Thus $m_{MTO,1} = 156490.7498$ kg.

Iteration is done until the difference of the new calculated and old maximum take-off mass is very small. Since the local mass growth is 1 kg, the global mass growth corresponds to the mass growth factor. The results of the further iteration steps are shown in the Table 3.2.

Iteration step	Maximum take-off [kg]	Difference [%]	Δm_G resp. k_{MGW}
1	156490.7498	74.9887%	1.75
2	156491.3122	32.1352%	2.31
3	156491.7339	18.2372%	2.73
4	156492.0501	11.5664%	3.05
5	156492.2872	7.7743%	3.29
6	156492.4651	5.4093%	3.47
7	156492.5984	3.8482%	3.60
8	156492.6984	2.7788%	3.70
9	156492.7734	2.0274%	3.77
10	156492.8296	1.4901%	3.83
11	156492.8718	1.1010%	3.87
12	156492.9034	0.8167%	3.90
13	156492.9271	0.6074%	3.93
14	156492.9449	0.4528%	3.94
15	156492.9582	0.3380%	3.96
16	156492.9682	0.2526%	3.97
17	156492.9757	0.1889%	3.98
18	156492.9813	0.1414%	3.98
19	156492.9855	0.1059%	3.99
20	156492.9887	0.0793%	3.99
21	156492.9911	0.0594%	3.99
22	156492.9929	0.0445%	3.99
23	156492.9942	0.0334%	3.99
24	156492.9952	0.0250%	4.00
25	156492.9959	0.0188%	4.00
26	156492.9965	0.0141%	4.00
27	156492.9969	0.0105%	4.00
28	156492.9972	0.0079%	4.00
29	156492.9975	0.0059%	4.00
30	156492.9977	0.0044%	4.00
31	156492.9978	0.0033%	4.00
32	156492.9979	0.0025%	4.00
33	156492.9980	0.0019%	4.00
34	156492.9980	0.0014%	4.00
35	156492.9981	0.0011%	4.00
36	156492.9981	0.0008%	4.00
37	156492.9981	0.0006%	4.00
38	156492.9981	0.0004%	4.00
39	156492.9981	0.0003%	4.00
40	156492.9982	0.0003%	4.00
41	156492.9982	0.0002%	4.00
42	156492.9982	0.0001%	4.00

 Table 3.2
 Further iteration steps and results

Thus the mass growth factor k_{MGW} for the Boeing 767-300 is approximately 4

3.3 Mass Growth Factor for 90% of Current Aircraft

From Robson 2019, the "World Airliner Census" is used here. This census data includes all commercial aircraft with jet and turboprop engine that are in operation worldwide or have been permanently contracted with airlines. Exceptions are those types that carry fewer than 14 passengers or equivalent cargo. This includes fleets of Western, Chinese and Russian/Ukrainian aircraft.

These aircraft figures are used to determine 90% of all current flying commercial aircraft. The evaluation can be found in Appendix A. For these aircraft the determination of the mass growth factor is required. In addition, this must be calculated for the supersonic passenger aircraft Concorde and TU-144. The calculation is done using the method described in Section 3.1 of this project.

Table 3.3 shows the results of the mass growth factor. The aircraft are arranged according to their current number. Thus, the Boeing 737-800 is the currently most operated aircraft. The two supersonic passenger aircraft Concorde and TU-144 are listed far below. In addition, the sources are listed from which the masses (maximum take-off mass, operating empty mass, maximum payload) for the respective aircraft have been taken.

The fuel mass shall be determined using Equation (3.13). The mass fractions are then determined from these masses to calculate the mass growth factor.

aircrait					
Aircraft	m _{MTO}	m _{OE}	m _{MPL}	k _{мсw}	Sources
	[kg]	[kg]	[kg]	Maw	
Boeing 737-800	78220	41480	14690	5.32	Jenkinson 2019
A320-200	73500	42100	18633	3.94	Jackson 2011
A320neo	79000	44300	20000	3.95	Airbus 2005, Wiki 2020
A321neo	97000	50100	25500	3.80	Airbus 2005a, Wiki 2020
A321-200	89000	48000	22780	3.90	Jenkinson 2019
A319-100	64000	39200	17390	3.68	Jenkinson 2019
Boeing 737-700	69400	37585	11610	5.97	Jenkinson 2019
ATR 72-500	22500	12950	7350	3.06	Jackson 2011
Boeing 777-300 ER	299370	155960	68570	4.36	Jackson 2011
Embraer 175	37500	21810	9890	3.79	Jackson 2011
Boeing 787-9	244940	128850	52587	4.65	Boeing 2018, Wiki 2020c
A330-300	217000	118189	48400	4.48	Jenkinson 2019
Boeing 767-300	156489	87135	39140	3.99	Jenkinson 2019
A350-900	280000	142400	53300	5.25	Airbus 2005b, Wiki 2020a
Boeing 757-200	115900	58040	25690	4.51	Jenkinson 2019
A330-200	230000	120200	36400	6.31	Jenkinson 2019
Boeing 737-900	74389	42901	19831	3.75	Boeing 2013, Wiki 2020e
DHC Dash 8-400	24993	14968	7257	3.44	Lambert 1991
Embraer 190	50300	28080	13530	3.71	Jackson 2011
Bombardier CRJ900	36500	21430	10320	3.53	AirlinesInform 2020
A220-100	63049	35221	15127	4.16	Airbus 2019, Wiki 2020d
Boeing 777-200	242670	135875	54635	4.44	Jenkinson 2019
Embraer 145	20600	11940	5160	3.99	Jackson 2011
Boeing 787-8	219550	108860	45439	4.83	Jackson 2011
Boeing 747-400	396830	181484	61186	6.48	Jenkinson 2019
Boeing 737-300	56470	31869	16030	3.52	Jenkinson 2019
A380-841	560000	270015	90985	6.15	Jackson 2011
Viking Air Twin Otter 400	5670	3121	1474	3.84	Jackson 2011
Bombardier CRJ700	33000	19730	8530	3.86	AirlinesInform 2020a
Boeing 737-400	62820	33370	17740	3.54	Jenkinson 2019
ATR42-500	18600	11250	5450	3.41	Jackson 2011
A330-800neo	251000	132000	44000	5.70	Wiki 2020b
Boeing MD-81	58061	35330	18195	3.19	Jenkinson 2019
Boeing 777F	347450	145150	102000	3.40	Jackson 2011
A300-600R	150000	79666	33300	4.50	Jenkinson 2019
Saab 340 B Plus	13150	8140	3880	3.38	AirlinesInform 2020b
Fairchild Metro/Merlin III	6577	3963	2214	2.97	Wiki 2020f
Beechcraft 1900D	7765	4732	1950	3.98	Zimex 2020, Wiki 2019
A350-1000	316000	155000	68000	4.64	Jackson 2011
Embraer 195	48790	28970	12720	3.83	Jackson 2011
Embraer 170	35000	20770	9000	3.00	Jackson 2011
	55770	211 4 0	2000	3.77	Juck5011 2011

 Table 3.3
 Evaluation of 90% of all current flying commercial aircraft including two supersonic aircraft

Boeing 787-10	254011	135500	57277	4.43	Boeing 2018, Wiki 2020c
Concorde (Supersonic)	187700	78700	12000	15.64	AirlinesInform 2020c
TU-144 (Supersonic)	180000	85000	15000	12	Jane 1982

4 Mass Growth Factor Beyond Iteration

4.1 Generalization of Iteration Results

By calculating the mass growth factor by iteration the following important findings can be made:

1. Larger aircraft do not necessarily have a higher mass growth factor.

The value of the factor depends primarily on the operating empty mass and the fuel mass fraction. If the maximum take-off mass is changed, but the mass fractions remain unchanged, this has no effect on the value of the mass growth factor. The size of the aircraft and thus the maximum take-off mass is not the decisive factor.

2. It does not matter how large the local mass growth is; the mass growth factor remains unaffected.

The maximum take-off mass converges against a value. The difference between the new and original take-off mass results in the global mass growth. This corresponds to a multiple (mass growth factor) of the local mass growth. Should mean:

$$\Delta m_G = k_{MGW} \cdot \Delta m_L \tag{4.1}$$

The local mass growth thus has a direct influence on the global mass growth and thus on the maximum take-off mass - but not on the mass growth factor.

3. With the mass growth factor, it is mainly a matter of increasing a fixed mass.

The fuel is not part of the fixed mass. The "first law of aircraft design" (3.2) including the growth in mass can be described as

$$m_{MTO} + \Delta m_G = m_{MPL} + m_{OE} + \Delta m_L + m_F \tag{4.2}$$

From this it can be deduced that when the aircraft takes off, it does not matter whether there is one kg more wing mass (operating empty mass) or one kg more payload on board. The mass growth factor for a growth in the operating empty mass is therefore the same factor as for a growth in the payload. 4. Values of the mass growth factor for different aircraft categories.

The mass growth factor for 90% of all current flying commercial aircraft was determined in Section 3.2 of this project. Based on these values the average mass growth factor for different aircraft categories has to be determined. The categories and the result for the mass growth factors are summarized in Table 4.1.

Table 4.1	Mass growth factor for different aircraft categories
-----------	--

	Aircraft categories			
	Wide-Body	Narrow-Body	Supersonic	Subsonic
k _{MGW}	4.91	3.85	4.23	13.82

Which aircraft have been grouped into which category can be seen in Appendix B.

5. Old long-range aircraft are more sensitive to local mass growth than new short-range aircraft.

To confirm this result, the aircraft from Section 3.2 are categorized into old long-range and new short-range aircraft. For the respective aircraft the mass growth factors are summarized in Figures 4.1 and 4.2.

The mass growth factor of 4.38 of the old long-range aircraft is on average higher. This is mainly due to the fact, that they fly long distances and require more fuel. This means increased fuel mass, which in turn leads to a large mass growth factor.



Figure 4.1 Mass growth factor of old long-range aircraft (Screenshot: Evaluation_90%_aircraft_+_category.xlsm)



 Figure 4.2
 Mass growth factor of new short-range aircraft (Screenshot: Evaluation_90%_aircraft_+_category.xlsm)

4.2 Mass Growth Factor from Operating Empty Mass and Fuel Mass Fraction

The mass growth factor can be determined using the operating empty mass and fuel mass fraction of an aircraft. This knowledge shall be used to give a typical value of the mass growth factor for the three types of ranges (long-range, medium-range and short-range aircraft).

For this purpose, a typical value for the mass fractions is first determined for the respective range. Typical values for the operating empty mass fraction are taken from the Figure 4.3.



Figure 4.3Operating empty mass fraction as a function of design range (according to Scholz
2015, Section 5, p.29)

From this, the following values are taken for the operating empty mass fraction:

- For long-range aircraft: 0.45
- For short-range aircraft: 0.6

For medium-range aircraft, the arithmetic mean of the two numbers is used. This would be 0.525.

To obtain typical values for the fuel mass fraction, Figure 4.4 is used.



Figure 4.4 Payload fraction as a function of maximum take-off mass (according to Scholz 2018, slide 30)

In this Figure 4.4 the payload fraction is plotted over the aircraft size. For short-range aircraft the payload fraction is given as 0.25. For long-range aircraft a value of 0.1 can be read, because long-range aircraft have a large take-off mass.

Using the Equation (4.3):

$$\frac{m_F}{m_{MTO}} = 1 - \frac{m_{PL}}{m_{MTO}} - \frac{m_{OE}}{m_{MTO}}$$
(4.3)

And with the values for the operating empty mass and payload fraction for short and long-range, the fuel mass fraction is obtained:

- For long-range aircraft: 0.45
- For short-range aircraft: 0.15

For medium-range aircraft, the arithmetic mean of the two numbers is used. This would be 0.3.

The typical values for both mass fractions are determined. Now the mass growth factor is calculated for these values. The results are shown in Table 4.2.

	Range type						
	Short-range	Medium-range	Long-range				
m _{OE} /m _{MTO}	0,6	0,525	0,45				
m_F/m_{MTO}	0,15	0,3	0,45				
k _{MGW}	4	5,71	10				

 Table 4.2
 Mass growth factor for typical mass fractions

With these results it can be determined that the greater the mass growth factor, the greater the range. This relationship is examined in more detail in Section 4.4.

4.3 Mass Growth Factor from Payload Fraction

When determining the mass growth factor for current aircraft, a correlation between the mass growth factor and the payload fraction m_{MPL}/m_{MTO} is noticeable.

As calculated in Section 3.1.1, the mass growth factor for the Boeing 767-300 is approximately 4. The payload fraction of this aircraft is 0.25. The reciprocal value of the payload fraction thus corresponds to the mass growth factor. This picture appears for all examined aircraft.

It is therefore assumed that the mass growth factor is calculated as follows:

$$k_{MGW} = \frac{1}{\frac{m_{MPL}}{m_{MTO}}} \tag{4.4}$$

An equation for direct calculation of the mass growth factor is derived and an attempt is made to confirm Equation (4.4).

The Equation (3.1) can be paraphrased as

$$1 = \frac{m_{MPL}}{m_{MTO}} + \frac{m_{OE}}{m_{MTO}} + \frac{m_F}{m_{MTO}}$$
(4.5)

The operating empty mass and fuel mass fraction are summarized as "zero passenger weight" m_{ZP} :

$$\frac{m_{ZP}}{m_{MTO}} = \frac{m_{OE}}{m_{MTO}} + \frac{m_F}{m_{MTO}}$$
(4.6)

By including m_{ZP}/m_{MTO} in Equation (3.1) follows:

$$m_{MTO} = m_{MPL} + \frac{m_{ZP}}{m_{MTO}} \cdot m_{MTO}$$
(4.7)

Taking into account the local and global growth in masses, the result is

$$m_{MTO} + \Delta m_G = m_{MPL} + \frac{m_{ZP}}{m_{MTO}} \cdot (m_{MTO} + \Delta m_G) + \Delta m_L$$
(4.8)

Now the context (4.4) is derived instead of iterating as in Section 3.1. For this purpose, first of all, the transformation is done so that the maximum payload is on one side:

$$m_{MTO} + \Delta m_G - \frac{m_{ZP}}{m_{MTO}} \cdot (m_{MTO} + \Delta m_G) = m_{MPL} + \Delta m_L$$
(4.9)

This can be described as:

$$(m_{MTO} + \Delta m_G) \cdot (1 - \frac{m_{ZP}}{m_{MTO}}) = m_{MPL} + \Delta m_L$$
(4.10)

The maximum payload is given as payload fraction. To do this, the entire equation is divided by m_{MTO} . The expression $1 - m_{ZP} / m_{MTO}$ remains unchanged:

$$(1 + \frac{\Delta m_G}{m_{MTO}}) \cdot (1 - \frac{m_{ZP}}{m_{MTO}}) = \frac{m_{MPL}}{m_{MTO}} + \frac{\Delta m_L}{m_{MTO}}$$
(4.11)

Forming and dividing by $\Delta m_L/m_{MTO}$ results:

$$\frac{\frac{\Delta m_G}{m_{MTO}}}{\frac{\Delta m_L}{m_{MTO}}} = \frac{\frac{\frac{m_{MPL}}{m_{MTO}} + \frac{\Delta m_L}{m_{MTO}}}{1 - \frac{m_{ZP}}{m_{MTO}}} - 1}{\frac{\Delta m_L}{m_{MTO}}}$$
(4.12)

The ratio on the left side of the equation corresponds to the mass growth factor k_{MGW} . The right side of this equation is simplified.

The mass growth factor is thus obtained according to the Equation (4.4) assumed above:

$$k_{MGW} = \frac{1}{\frac{m_{MPL}}{m_{MTO}}}$$

The simplification of Equation (4.12) to Equation (4.4) is given in Appendix C.

This has confirmed the assumption that the reciprocal value of the maximum payload fraction corresponds to the mass growth factor. In addition, a further simple equation is given in which the mass growth factor can be calculated directly by applying the operating empty mass and fuel fraction.

The Equation (4.4) can be paraphrased as:

$$k_{MGW} = \frac{m_{MTO}}{m_{MPL}} \tag{4.13}$$

Forming from the "first law of aircraft design" (3.2) to m_{MTO}/m_{MPL} results:

$$k_{MGW} = \frac{m_{MTO}}{m_{MPL}} = \frac{1}{1 - \frac{m_F}{m_{MTO}} - \frac{m_{OE}}{m_{MTO}}}$$
(4.14)

Depending on the available mass fraction, the mass growth factor can be calculated directly using Equation (4.4) or (4.14).

4.4 Mass Growth Factor Depending on Technology and Range Requirements

In this section, the extent to which the mass growth factor is influenced by the technology and the requirement will be examined. The Breguet factor B is considered to check the dependence on the technology. This includes both, the maximum glide ratio E as well as the specific fuel consumption c as technology parameters. One of the important requirements is the range R.

In the previous section it was derived that the mass growth factor can be determined by the mass fractions. Equation (4.14) is used for the sensitivity test. Initially, functions of the operating empty mass and fuel mass fraction are determined depending on the technology parameter B and requirement R.

The project work from Lehnert (2018) provides several statistically based formulas for the determination of the operating empty mass fraction. One formula represents the operating empty mass fraction as a function of the range:

$$\frac{m_{OE}}{m_{MTO}} = 0.5967 - 0.00000166 \cdot R \tag{4.15}$$

A function is to be derived for the fuel mass fraction, which is dependent on both, the Breguet factor and the range. The "Breguet range equation" according to Scholz 2012 is used for the derivation:

$$R = \frac{E \cdot v}{c \cdot g} \ln \frac{m_1}{m_2} \tag{4.16}$$

In this case, m_1 is the take-off mass. For the landing mass m_2 applies:

$$m_2 = m_1 - m_F \tag{4.17}$$

The part before the logarithm of Equation (4.16) is the so-called Breguet factor B:

$$B = \frac{E \cdot V}{c \cdot g} \tag{4.18}$$

Thus Equation (4.16) can be paraphrased as:

$$R = B \ln \frac{m_1}{m_1 - m_F}$$
(4.19)

With $m_1 = m_{MTO}$ results:

$$R = B \ln \frac{1}{1 - \frac{m_F}{m_{MTO}}}$$
(4.20)

Dissolution according to the fuel mass fraction:

$$\frac{m_F}{m_{MTO}} = 1 - e^{-\frac{R}{B}}$$
(4.21)

There is now a function for the operating empty mass fraction, depending on range, and a function for the fuel mass fraction, depending on range and the Breguet factor. Inserting these two functions into Equation (4.14) gives:

$$k_{MGW} = \frac{1}{1 - (0.5967 - 0.00000166 \cdot R) - \left(1 - e^{-\frac{R}{B}}\right)}$$
(4.22)

This formula is used for the sensitivity test. The aircraft used is the A320-200. For this aircraft the following values from Table 4.3 are required.

Parameters		Value (unit)	Source
Cruising speed	Vcr	230 m/s	De Grave 2017
Range (at maximum payload)	R	1600 NM resp. 2963200 m	Jackson 2011
Glide ratio in cruise flight	Ε	17.91	De Grave 2017
Specific fuel consumption	С	1.63 E-05 kg/N/s	De Grave 2017

 Table 4.3
 Sensitivity Test - Parameters for the A320-200

4.5 Mass Growth Factor Sensitivity with Range

When checking the dependence of the mass growth factor on the range, the Breguet factor is calculated using the values in Table 4.3 and is not changed thereafter. For the A320-200, this gives a Breguet factor value according to Equation (4.18) of B = 25761242.8 m.

At a range of 1600 NM and using Equation (4.22), a mass growth factor of $k_{MGW} = 3.3636$ results. In Section 3.2 a factor of 3.9446 was determined for the A320-200. This was done using the masses that were researched for the finished aircraft.

In this case, however, the mass fractions are determined with a statistically based formula and by using dimensioning approaches with estimated values for the glide ratio and specific fuel consumption. The deviating mass growth factor is not critical in this context, as it is mainly a matter of sensitivity.

For the sensitivity test the range of 1600 NM is increased and the change of the mass growth factor is recorded. The results are summarized in Table 4.4.

Panga [NM]	Change in range	1.	Effect on	Effect on original addi-
Kange [INM]	[%]	R _{MGW}	k _{MGW} [%]	tional kg [%]
1600	0	3.36	0.00	
1616	1	3.37	0.34	1.13
1632	2	3.38	0.68	2.27
1760	10	3.47	3.45	11.62
1920	20	3.60	7.11	23.92

Table 4.4 Results of sensitivity test depending on the requirement

If the range is changed by 1%, the mass growth factor changes by 0.34%. If the range growths by 10%, the factor growths by 3.45%. It should be noted, however, that the sensitivity depends very much on the initial range at maximum payload. If the range is twice as large, the sensitivities will be higher.

In the right column, the effect is shown on the original additional kg. This shows by how much more the aircraft is loaded. A local mass growth of 1 kg (original additional kg) results in a factor of 3.3636. This means that the local mass growth of 1 kg leads to a global mass growth of 3.3636 kg according to Equation (4.1).

By turning the range adjustment screw, the mass growth factor changes. In this case to 3.3749, if the range is increased by 1%. This again means a global mass growth of 3.3749 kg.

If one look at how much the global mass growth has risen, you can see:

3.3749 kg - 3.3636 kg = 0.0113 kg

This value in relation to the original additional kg corresponds to 1.13%.

Furthermore, it can be concluded from this study, that the more the range growths in size, the greater the mass growth factor. Aircraft with such a longer range transport more fuel, which growths its fraction and thus the mass growth factor is correspondingly high. The fraction of fuel growths with the range, and the operating empty mass fraction decreases slightly. In sum, the two mass fractions go towards 1.

Equation (4.14) can also be paraphrased as:

$$k_{MGW} = \frac{1}{1 - \left(\frac{m_F}{m_{MTO}} + \frac{m_{OE}}{m_{MTO}}\right)}$$
(4.23)

The more the sum of the two mass fractions approaches 1, the more the mass growth factor growths. A further range can only be reached up to the point where $k_{MGW} = \infty$ is. Figure 4.5 shows how the mass fractions and the mass growth factor behave with increasing range.



Figure 4.5 Mass growth factor and mass fraction over the range (Screenshot: Sensitivity_with_Range.xlsm)

4.6 Mass Growth Factor Sensitivity with Technology

When checking the dependence of the mass growth factor on technology, the range is kept constant and the Breguet factor is changed. The range in this case is 1600 NM. We begin with the Breguet factor of B = 25761242.8 m. The mass growth factor therefore has a value of $k_{MGW} = 3.3636$.

By changing the Breguet factor and the resulting effect on the mass growth factor, the sensitivity of this factor as a function of the glide ratio E or the specific fuel consumption c can be closed. If the Breguet factor is increased by 1%, this corresponds to a 1% growth in glide ratio or a 1% reduction in specific fuel consumption.

For the sensitivity test, the Breguet factor of B = 25761242.8 m is increased and the change in the mass growth factor was recorded. The results are summarized in Table 4.5.

Breguet factor	Change in Breguet	1-	Effect on	Effect on original addi-
[m]	factor [%]	K _{MGW}	k _{MGW} [%]	tional kg [%]
25761242.75	0	3.36	0.00	
26018855.18	1	3.35	0.34	1.15
26276467.61	2	3.34	0.67	2.26
28337367.03	10	3.26	3.06	10.28
30913491.30	20	3.17	5.48	18.45

 Table 4.5
 Results of sensitivity test depending on the technology

If the Breguet factor is changed by 1%, the mass growth factor changes by 0.34%. For a 10% conversion, it is 3.06%. The right-hand column again shows the effect on the original additional kg.

If the Breguet factor is increased, the mass growth factor becomes smaller. A growth in the Breguet factor means an improvement and thus a removal from the critical state. The sum of the operating empty mass and fuel mass fraction in Equation (4.23) thus moves further and further away from 1.

A deterioration or reduction in the Breguet factor can be obtained by increasing the range, which in turn will lead to a growth in the mass growth factor. The Table 4.6 shows the results obtained by reducing the Breguet factor.

	······································				
Breguet factor	Change in Breguet	1,	Effect on	Effect on original addi-	
[m]	factor [%]	ĸ _{MGW}	k _{MGW} [%]	tional kg [%]	
25761242.75	0	3.36	0.00		
25503630.33	1	3.37	0.35	1.18	
25246017.90	2	3.38	0.71	2.38	
23185118.48	10	3.49	3.96	13.31	
20608994.20	20	3.67	9.29	31.24	

 Table 4.6
 Results of sensitivity test depending on technology - with reduction

Comparing the effects in terms of increasing the mass growth factor and the original additional kg, it is striking that the Breguet factor has a greater effect than increasing the range, at least on this aircraft. In any case, it can be said that the mass growth factor is greater with a higher specific fuel consumption and a lower glide ratio.

4.7 Mass Growth Factor as an Economical Indicator

The mass growth factor as the reciprocal of the ratio of maximum payload to maximum takeoff mass can also be considered as a kind of economic efficiency factor.

According to Scholz 2015, the Direct Operation Costs (DOC) are composed of:

$$C_{DOC} = C_{DEP} + C_{INT} + C_{INS} + C_F + C_M + C_C + C_{FEE}$$
(4.24)

The take-off mass includes the material that makes up the aircraft price and fuel. The depreciation is related to the price and this in turn is related to the kg aircraft. Therefore the operating empty mass corresponds to the depreciation. The fuel mass is the same as the fuel and the requirement is included as payload.

Maintenance costs are not considered. Thus, both fixed and variable costs are included. Thus the maximum take-off mass can be assumed approximately as a substitute value for the DOC:

$$DOC \approx m_{MTO} = m_{OE} + m_F + m_{PL} \tag{4.25}$$

According to Scholz 2015, the DOC can be viewed in different ways. The DOC can be considered as costs "which are incurred for an aircraft (aircraft, index: a/c) of a fleet within one year (annual costs, index: a). They are annual aircraft annual costs". (Scholz 2015, Section 14, p.18):

$$C_{DOC} = C_{a/c,a} \tag{4.26}$$

According to Scholz 2015, the aircraft trip costs $C_{a/c,t}$ are obtained, when the DOC are calculated for a defined flight (flight,cycle or trip, index: t):

$$C_{a/c,t} = \frac{C_{a/c,a}}{R \cdot n_{t,a}} \tag{4.27}$$

Here is $n_{t,a}$ the number of flights an aircraft can perform in a flight mission (number of flights per year) and *R* is the range.

The seat-mile costs are the result, when the DOC are based on "the range *R* and the number of seats n_s or the maximum number of passengers n_{PAX} ." (Scholz 2015, Section 14, p.18):

$$C_{s,m} = \frac{C_{a/c,t}}{R \cdot n_{PAX}} = \frac{C_{a/c,a}}{R \cdot n_s \cdot n_{t,a}}$$
(4.28)

Except for the cargo, the payload is proportional to the number of seats. The number of flights depends on how efficiently the aircraft can be operated. The range is given by the specification, among other things. The denominator therefore acts as a substitute value for the maximum payload. The denominator also contains the DOC, which is almost equal to the maximum take-off mass.

Thus follows:

$$\frac{DOC}{n_{t,a}} \approx \frac{m_{MTO}}{m_{PL}} = \frac{1}{\frac{m_{PL}}{m_{MTO}}}$$
(4.29)

The mass growth factor is thus related to the DOC per seat (seat costs). If the aircraft is able to transport many people who share the costs, the aircraft has a large payload. Therefore it has a small mass growth factor and is efficient.

If one wants to use the mass growth factor to consider a comparison of aircraft with regard to their efficiency effect, it should be noted that they must have the same range and the same cruising speed. The cruising speed has a direct effect on the number of flights per year. This is explained in the next section.

4.8 The Problem of a Large Mass Growth Factor

At this point it is interesting to look at the Concorde. This one has a mass growth factor of nearly 15. With this aircraft one found out that there is a DOC problem. However, it was assumed that due to its speed more tickets are sold according to the following motto: If the aircraft flies twice as fast, it can also sell twice as many tickets.

There were two approaches to solving the transport problem at that time. Either they built a large aircraft, the Boeing 747, or a fast aircraft, the Concorde. In the beginning it was not obvious that the Boeing 747 was more economical. It was assumed that both would be equally profitable and that the Concorde would also be much faster. For this, however, the same mass growth factors would have had to be available. The Concorde has a much larger mass growth factor than the Boeing 747 and is therefore able to transport only a few passengers.

The very high speed of the Concorde had a negative effect in the end. To estimate the seat-milecosts after Equation (4.28) one needs the number of flights per year. This gives it after Scholz 2015 with:

$$n_{t,a} = \frac{U_{a,f}}{t_f} \tag{4.30}$$

Here is $U_{a,f}$ the annual aircraft usage and t_f the defined aircraft time, which according to Scholz 2015 results from the ratio of range to cruising speed:

$$t_f = \frac{R}{V_{CR}} \tag{4.31}$$

So if you fly faster, the defined aircraft time is smaller. A smaller flight time means a high number of flights per year. This in turn means that the seat-mile costs are significantly lower when flying faster for the same cost. The costs will also growth significantly, because the flight continues at higher speed in the same available time (variable costs), whereas the fixed costs (depreciation, insurance, etc.) are independent of the route and are better distributed over the increased number of kilometers when the flight continues.

On the basis of the Concorde, it can thus be stated that an excessively large mass growth factor is disastrous. To support this statement a quotation is quoted, which has dealt with the Concorde in detail.

"What happened is that it [Concorde] had to be redesigned twice, perhaps two and half times, completely because of weight growth, and in the course of redesigning it, more expensive methods of construction had to be adopted and more expensive materials, and the result was that the production cost turned out to be about four times as high as had been anticipated. The economics of it went completely. They had never been very good, but it became obvious from the mid-1960s to anybody that this was economically disastrous, even there were doubts about whether it was technically possible which still had to be resolved." (Glancey 2015).

5 Discussion

Various definitions of the mass growth factor are given. One of these definitions accurately reflects the mass growth factor. Accordingly, the factor quantifies how much the take-off mass is increased, caused by a local mass growth. In the present work two methods for the determination of the mass growth factor were derived.

The occurrence of a mass snowball effect can be clearly shown by the first method by iteration. A mass growth iteration is shown. Thus a convergence of the take-off mass increase takes place when the mass is increased locally. It has to be considered, that the length of the iteration can vary from aircraft to aircraft until the maximum take-off mass reaches the limit.

This iteration method shows, that there is no dependence of the mass growth factor on the size of the aircraft. So there is a contradiction to the opinion of the book Jenkinson 1999, which made the assumption, that a bigger aircraft has a higher factor. The comparison of two Boeing aircraft of Section 3.2 confirms the findings of this work. While the mass growth factor of the Boeing 737-700 is 6, the much larger Boeing 787-9 has a factor of 4.6.

The factor size is bound to the mass fractions. Aircraft with a larger operating empty mass and fuel mass fraction and consequently a lower payload fraction have a higher mass growth factor. From the SAWE paper Saelman 1973 it is already evident that a high mass growth factor exists with a small payload fraction. The second method for determining the mass growth factor confirms this correlation.

By the second method a simple Equation (4.4) resp. (4.14) is provided to calculate the mass growth factor directly without iteration. Therefore the factor can be determined easily by the payload fraction or the operating empty mass and fuel mass fraction. Comparable formulas can already be found in Ballhaus 1954, Fürst 1999 and SAWE 2019. The formulas from the literature also include an additional subdivision of the operating empty mass. A formula derivation is not available for Fürst 1999 and SAWE 2019. The derivation cannot be fully understood in Ballhaus 1954, which is not the case in this project work.

At Roskam 1989, the mass growth factor for an increase in operating empty mass and payload is looked at. However, the mass growth factor for an increase in payload is the same factor as for an increase in operating empty mass. The decisive factor is the increase of a fixed mass. A distinction whether one kg of additional weight is in the wing mass (operating empty mass) or in the payload is not important for aircraft take-off. The two SAWE papers Ballhaus 1954 and Saelman 1973 already contain this statement.

The different literature sources also contain values for the mass growth factor. Exactly which aircraft were used to determine these values remains unknown. Furthermore it can be seen that there is a difference between the values of the factor in the literature and the values calculated

in this work. The results of this work do not confirm the values at Saelman 1973, Fürst 1999 and Müller 2003. For an aircraft with an operating empty mass of 0.5, a factor value of 1.3 was calculated with the data from Torenbeek 1988. For numerous aircraft mentioned in Chapter 3.2, the operating empty mass fraction is approximately 0.5, but the mass growth factor is not as low for any of these aircraft.

In the factor analysis of technology and requirement the following can be determined: Aircraft with higher specific fuel consumption and lower glide ratio have a higher mass growth factor. With a longer range, the mass growth factor increases significantly. Already in the paper Ballhaus 1954 it was shown how much the factor depends on the range. Of great importance is the fact that at a certain range the mass growth factor develops towards infinity. The reason for this is again that at a long-range and consequently a large fuel mass, almost nothing is left of the 1 in the denominator of formula (4.14).

Furthermore, the mass growth factor provides information about the DOC per seat (seat costs). An aircraft with a low mass growth factor has a higher efficiency and economy. The aircraft can therefore transport a large number of people who together bear the costs. The ATR 72 has a factor of only 3.1 and convinces with its good performance among the aircraft sold in large numbers. In contrast, the Concorde, which has a factor of almost 15, can be described as an "economic disaster".

6 Summary and Recommendations

The mass snowball effect is the iterative interaction between the take-off mass on the one hand and the component masses of the system, structure or drive on the other. If the operating empty mass of the aircraft increases, fuel consumption also increases. As a result, larger fuel tanks become necessary, which primarily leads to an increase in the maximum take-off mass, and structural component adjustments are also necessary. The resulting increase in component masses results in a higher fuel mass and thus a higher take-off mass. This results in an iterative process.

The so-called mass growth factor quantifies this effect. It quantifies the extent of the increase in take-off mass that results from a local increase in mass. In the course of some phases of aircraft design, the mass increases rather than decreases compared to the original estimates. Therefore the factor is called mass growth factor. From a mathematical point of view, however, there is no difference to a mass reduction factor.

Based on literature research, a presentation of first findings as well as investigations on mass growth and mass growth factor is made. If the operating empty mass has to be increased, the design draft also has to be revised, which in turn has an effect on the take-off mass. If the takeoff mass varies, it must be accepted that the aircraft performance will be restricted. In the course of the literature research, an explanation of the methods and formulas is also provided. The values of the mass growth factor are also given; this is used to determine a reference value for the factor.

By means of the "first law of aircraft design" and by iteration, a method for determining the mass growth factor is developed. Here the mass growth factor represents the ratio of the increase of the global mass (take-off mass) by the arbitrary increase of the local mass (empty mass). In this method an iteration of the mass growth is represented, where there is a convergence of the take-off mass increase with local mass increase. The lowest mass growth factor that is possible is 1. From this it follows that in case of an empty mass increase there is no need for further measures such as structural strengthening.

The mass growth factor is calculated with this method for such a large number of renowned passenger aircraft that it covers 90% of the passenger aircraft currently flying. If these aircraft are categorized, the value of the mass growth factor can be summarized as follows: Subsonic passenger aircraft have an average mass growth factor of 4.2, narrow-body planes of 3.9 and wide-body planes (whose routes tend to be longer) of 4.9, whereas the factor for supersonic passenger aircraft is approximately 14.

It can be stated that the mass growth factor for some aircraft is the same as the reciprocal of the payload fraction. This picture recurs for all aircraft. With this knowledge the derivation of an equation is done, with which the factor can be calculated directly (without iteration).

The mass growth factor is analyzed in more detail using the two methods of determination iteration method and equation. It can be seen that the value of the mass growth factor does not depend on the size of the aircraft, but on the fractions of the masses. Aircraft with a larger operating empty mass and fuel mass fraction and a smaller payload fraction have a higher mass growth factor.

The decisive factor is the increase of a fixed mass, which makes it the same for the payload and the operating empty mass. If one examines to what extent the mass growth factor depends on requirements and technology, it can be seen that a higher specific fuel consumption and a lower glide ratio is accompanied by a larger factor. With increasing range, the factor also increases considerably.

Furthermore, there is a correlation between the mass growth factor and the DOC per seat. If the aircraft is able to transport a large number of people, who together bear the costs, the payload fraction of this aircraft is high. The mass growth factor of this aircraft is therefore low and therefore the aircraft is efficient. The negative example of the Concorde shows, how a too high mass growth factor can have a considerable effect on the economic efficiency.

Since the present project work focused on the determination of the mass growth factor, there was no room for a more detailed investigation of the location of the local mass increase. However, this aspect could be a relevant approach for further investigations.

The decisive factor is where on the aircraft the mass growth takes place. If the mass increases at the wing, this is not of high importance, because there the mass is carried equally. Engines on the wing can counteract the increase of the bending moment. However, this is different if the mass is increased locally in the front area of the aircraft. In this case, assuming the completion of the aircraft construction, an increase in the production of downforce must take place. Therefore a variation of the factor with the weight increase location along the fuselage station is possible.

List of References

 AIRBUS, 2005. A320 Aircraft Characteristics – Airport and Maintenance Planning. Blagnac, France: Airbus S.A.S.
 Available from: <u>https://bit.ly/2IjsDUY</u>
 Archived as: <u>https://perma.cc/S8XL-P92S</u>

 AIRBUS, 2005a. A321 Aircraft Characteristics – Airport and Maintenance Planning. Blagnac,

 France: Airbus S.A.S.

 Available from: https://bit.ly/30hiodz

 Archived as:
 https://perma.cc/YW2P-7TSW

 AIRBUS, 2005b. A350 Aircraft Characteristics – Airport and Maintenance Planning. Blagnac,

 France: Airbus S.A.S.

 Available from: https://bit.ly/3jezUqn

 Archived as:
 https://perma.cc/XA3A-WH5N

AIRBUS, 2019. Airbus announces major performance improvement to its latest single-aisle aircraft – the A220 Family. France: Airbus S.A.S.
 Available from: <u>https://bit.ly/30j3hk2</u>
 Archived as: <u>https://perma.cc/NF4S-UHY8</u>

AIRLINESINFORM, 2020. *Bombardier CRJ-900*. Russia: Aminta. Available from: <u>https://bit.ly/2G9XyWH</u> Archived as: <u>https://perma.cc/5EEV-ERGC</u>

AIRLINESINFORM, 2020a. Bombardier CRJ-700. Russia: Aminta.

Available from:https://bit.ly/3jfHKzZArchived as:https://perma.cc/RYG6-BZ5A

AIRLINESINFORM, 2020b. *Saab 340*. Russia: Aminta. Available from: <u>https://bit.ly/345JyW9</u> Archived as: <u>https://perma.cc/BJK3-N2LD</u>

AIRLINESINFORM, 2020c. *Concorde*. Russia: Aminta. Available from: <u>https://bit.ly/2Gcq2PD</u> Archived as: https://perma.cc/UPZ7-3XFF

 BALLHAUS, William F., 1954. Clear Design Thinking Using the Aircraft Growth Factor. In: 14th National Conference, Hilton Hotel, Fort Worth, Texas, May 2-5. Los Angeles, California: Society of Allied Weight Engineers, Inc. Avaiable from: https://www.sawe.org/papers/0113 BOEING, 2013. Boeing 737 Aircraft Characteristics for Airport Planning. Seattle, Washington: Boeing Commercial Airplanes
 Available from: <u>https://bit.ly/2GdT1SV</u>
 Archived as: <u>https://perma.cc/VXK7-WNJZ</u>

BOEING, 2018. Boeing 787 Aircraft Characteristics for Airport Planning. Seattle, Washington: Boeing Commercial Airplanes
 Available from: <u>https://bit.ly/3ieXkKU</u>
 Archived as: <u>https://perma.cc/B2ZJ-EQS3</u>

COLLINS, 2020. *payload*. Glasgow: CollinsDictionary.com Available from: <u>https://bit.ly/34b5YFd</u> Archived as: <u>https://perma.cc/3XA5-SXL9</u>

DE GRAVE, Emiel, 2017. *Reverse Engineering of Passenger Jet Classified Design Parameters*. Hamburg, Germany: Department of Automotive and Aeronautical Engineering, University of Applied Science of Hamburg, Master Thesis. Available from: <u>https://nbn-resolving.org/urn:nbn:de:gbv:18302-aero2017-08-25.017</u>

- DIN, 9020-5, 1992. *Luft- und Raumfahrt; Masseaufteilung für Luftfahrzeuge schwerer als Luft.* Berlin: Deutsches Institut für Normung e. V.
- DUDEN, 2019. Sensitivität, die. Bibliographisches Institut GmbHAvaiable from:https://www.duden.de/rechtschreibung/SensitivitaetArchived as:https://perma.cc/YHG3-VH2R
- FÜRST, A., 1999. *Masseeinfluß-/Massezuwachsfaktor (growth-factor)*. Ottobrunn, Germany: IABG LTH-Koordinierungsstelle, Aeronautical manual.
- GLANCEY, Jonathan, 2015. *Concorde: The Rise and Fall of the Supersonic Airline*. London, England: Atlantic Books.
- HOWE, Denis, 2000. *Aircraft Conceptual Design Synthesis*. London: Professional Engineering Publishing.

JACKSON, Paul, 2011. Jane's All the Worlds Aircraft 2010-11. UK: MPG Books Group.

JENKINSON, Lloyd; SIMKIN, Paul; RHODES, Darren, 1999. *Civil Jet Aircraft Design*. Oxford, UK: Butterworth-Heinemann.

- JENKINSON, Lloyd; SIMKIN, Paul; RHODES, Darren, 2019. Civil Jet Aircraft Design Aircraft Data File. Oxford, UK: Butterworth-Heinemann.
 Avaiable from: <u>https://bit.ly/2GhxWac</u>
 Archived as: <u>https://perma.cc/4H5W-ZT5H</u>
- LAMBERT, Mark, 1991. Jane's All the Worlds Aircraft 1991-92. London: Butler & Tanner Limited.
- LEHNERT, Jan, 2018. Methoden zur Ermittlung des Betriebsleermassenanteils im Flugzeugentwurf. Hamburg, Germany: Department of Automotive and Aeronautical Engineering, University of Applied Science of Hamburg, Master Thesis.
 Availkable from: <u>https://nbn-resolving.org/urn:nbn:de:gbv:18302-aero2018-05-24.014</u>
- MÜLLER, Friedrich, 2003. Flugzeugentwurf : Entwurfssystematik, Aerodynamik, Flugmechanik und Auslegungsparameter für kleinere Flugzeuge. Fürstenfeldbruck: TFT-Verlag.
- RISSE, Kristof, 2016. *Preliminary Overall Aircraft Design with Hyprid Laminar Flow Control*. Aachen, Germany: Faculty of Mechanical Engineering, RWTH Aachen, Dissertation. Available from: <u>https://doi.org/10.18154/RWTH-2017-00974</u>
- ROBSON, Melanie, 2019. *Flight Magazine: Perfect formation*. Sutton: Flight Global, 2019-07-30.
- ROSKAM, Jan, 1989. *Aircraft Design. Vol.1: Preliminary Sizing of Aircraft*. Ottawa, Kansas: Design Analysis and Research Corporation.
- SAELMAN, B., 1973. *The Growth Factor Concept*. In: 32nd Annual Conference, London, England, June 25-27. Los Angeles, California: Society of Allied Weight Engineers, Inc. Available from: <u>https://www.sawe.org/papers/0952</u>
- SAWE, 2019. Introduction to Aircraft Weight Engineering. Los Angeles, California: Society of Allied Weight Engineers, Inc.
 Available from: <u>https://www.sawe.org/technical/publications/aircrafttextbook</u>
- SCHOLZ, Dieter, 2012. Flugmechanik 1: Unterlagen zur Vorlseung. Hamburg, Germany: Department of Automotive and Aeronautical Engineering, University of Applied Science of Hamburg, Lecture Notes. Available from: https://bit.ly/2IrXoIL (Password required)

SCHOLZ, Dieter, 2015. *Aircraft Design.* Hamburg, Germany: Department of Automotive and Aeronautical Engineering, University of Applied Science of Hamburg, Lecture Notes. Available from: http://HOOU.ProfScholz.de

- SCHOLZ, Dieter, 2018. Evaluating Aircraft with Electric and Hybrid Propulsion. Hamburg, Germany: Department of Automotive and Aeronautical Engineering, University of Applied Science of Hamburg, Presentation. Available from: https://bit.ly/2GhLJxt
- SINKE, Jos, 2019. *How Aircraft Fly*. Delft, Netherlands: Department of Aerospace Engineering, Delft University of Technology, Lecture Notes.
 Available from: <u>https://bit.ly/3jakseW</u>
 Archived as: <u>https://perma.cc/NKX9-WKLY</u>

SKYbrary, 2020. Fuel. Brussels, Belgium: Eurocontrol.Available from:https://www.skybrary.aero/index.php/FuelArchived as:https://perma.cc/BB4R-4C54

- TAYLOR, John W. R., 1900. Jane's All the Worlds Aircraft 1989-90. California: Jane's Information Group.
- TORENBEEK, Egbert, 1988. Synthesis of Subsonic Aircraft Design. Delft: Delft University Press.
- WIKIPEDIA, 2018. Snowball effect.

Available from:https://en.wikipedia.org/wiki/Snowball_effectArchived as:https://perma.cc/4XTP-9J72

WIKIPEDIA, 2019. Beechcraft 1900.

 Available from:
 <u>https://en.wikipedia.org/wiki/Beechcraft_1900</u>

 Archived as:
 <u>https://perma.cc/7LSQ-GWGZ</u>

WIKIPEDIA, 2020. Airbus A320neo family.

Available from:https://en.wikipedia.org/wiki/Airbus_A320neo_family#SpecificationsArchived as:https://perma.cc/U3UG-R55K

WIKIPEDIA, 2020a. Airbus A350 XWB.

Available from:https://en.wikipedia.org/wiki/Airbus_A350_XWB#SpecificationsArchived as:https://perma.cc/R2FJ-5CUL

WIKIPEDIA, 2020b. Airbus A330neo.

Available from:https://en.wikipedia.org/wiki/Airbus_A330neo#cite_note-112Archived as:https://perma.cc/N27P-6XQY

WIKIPEDIA, 2020c. Boeing 787 Dreamliner.

Available from:https://en.wikipedia.org/wiki/Boeing_787_Dreamliner#SpecificationsArchived as:https://perma.cc/FQT3-HJ7M

WIKIPEDIA, 2020d. *Airbus A220*. Available from: <u>https://bit.ly/36j0paw</u> Archived as: <u>https://perma.cc/UF4Q-7WB4</u>

WIKIPEDIA, 2020e. Boeing 737.

Available from:https://en.wikipedia.org/wiki/Boeing_737Archived as:https://perma.cc/GP6D-2LUL

WIKIPEDIA, 2020f. *Fairchild Swearingen Metroliner*. Available from: <u>https://bit.ly/2S7SDIr</u> Archived as: <u>https://perma.cc/WXU7-YP24</u>

ZIMEX, 2020. Beechcraft 1900 Beechliner. Glattbrugg, Switzerland: Zimex Aviation Ltd.
 Available from: <u>https://bit.ly/3cG4ZAZ</u>
 Archived as: <u>https://perma.cc/YS9R-H24K</u>

All online references were accessed from the internet on or after 2019-12-10.

Appendix A

Aircraft Making Up 90% of the World Fleet

Aircraft Type	Number in Operation	Percent of total	Sum of most aircraft types in percent
Boeing 737-800	4804	16.83%	16.83%
A320	4135	14.48%	31.31%
A320neo	658	2.30%	33.61%
A321neo	160	0.56%	34.18%
A321	1650	5.78%	39.95%
Boeing 737 Max 8		0.00%	39.95%
A319	1249	4.37%	44.33%
Boeing 737-700	1005	3.52%	47.85%
ATR72	775	2.71%	50.56%
Boeing 777-300(ER)	829	2.90%	53.47%
Embraer 175	595	2.08%	55.55%
Boeing 787-9	451	1.58%	57.13%
A330-300	707	2.48%	59.61%
Boeing 767-300	622	2.18%	61.79%
A350-900	261	0.91%	62.70%
Boeing 757-200	600	2.10%	64.80%
A330-200	547	1.92%	66.72%
Boeing 737-900	550	1.93%	68.64%
De Havilland Canada Dash 8-400	502	1.76%	70.40%
Embraer 190	495	1.73%	72.14%
Boeing 737 Max TBD		0.00%	72.14%
Bombardier CRJ900	444	1.56%	73.69%
A220	77	0.27%	73.96%
Boeing 777-200	431	1.51%	75.47%
Embraer ERJ-145	422	1.48%	76.95%
Boeing 737 Max 10		0.00%	76.95%
Boeing 787-8	328	1.15%	78.10%
Boeing 747-400	331	1.16%	79.26%
Boeing 737-300	297	1.04%	80.30%
A380	233	0.82%	81.11%
Viking Air Twin Otter Total	263	0.92%	82.04%
Bombardier CRJ700	266	0.93%	82.97%
Boeing 737-400	261	0.91%	83.88%
ATR42	213	0.75%	84.63%
A330neo	16	0.06%	84.68%
Boeing MD-80	222	0.78%	85.46%
Boeing 777F	164	0.57%	86.04%
A300	202	0.71%	86.74%
Saab 340	202	0.71%	87.45%
Fairchild Metro/Merlin	196	0.69%	88.14%
Beechcraft 1900D	184	0.64%	88.78%
A350-1000	21	0.07%	88.85%
Embraer 195	160	0.56%	89.42%
Embraer 170	160	0.56%	89.98%
Boeing 787-10	29	0.10%	90.08%
Aircraft total	28550		

Table A.190% of current aircraft (according to Robson 2019)

Appendix B

Mass Growth Factor for Four Aircraft Categories

62

Wide-Body	Narrow-Body	Subsonic	Supersonio
Boeing 777-300 ER	Boeing 737-800	Boeing 737-800	Concorde
Boeing 787-9	A320-200	A320-200	TU-144
A330-300	A320neo	A320neo	
Boeing 767-300	A321neo	A321neo	
A350-900	A321-200	A321-200	
A330-200	A319-100	A319-100	
Boeing 777-200	Boeing 737-700	Boeing 737-700	
Boeing 787-8	ATR 72-500	ATR 72-500	
Boeing 747-100	Embraer 175	Boeing 777-300 ER	
A380-841	Boeing 757-200	Embraer 175	
A330-800neo	Boeing 737-900	Boeing 787-9	
Boeing 777F	De Havilland Canada Dash 8-400	A330-300	
A300-600R	Embraer 190	Boeing 767-300	
A350-1000	Bombardier CRJ900	A350-900	
Boeing 787-10	A220-100	Boeing 757-200	
	Embraer 145 ER	A330-200	
	Boeing 737-300	Boeing 737-900	
	Viking Air Twin Otter Total 400	De Havilland Canada Dash 8-400	
	Bombardier CRJ700	Embraer 190	
	Boeing 737-400	Bombardier CRJ900	
	ATR42	A220-100	
	Boeing MD-81	Boeing 777-200	
	Saab 340 B Plus	Embraer 145 ER	
	Fairchild Metro/Merlin	Boeing 787-8	
	Beechcraft 1900D	Boeing 747-400	
	Embraer 195	Boeing 737-300	
	Embraer 170	A380 [A380-841]	
		Viking Air Twin Otter Total 400	
		Bombardier CRJ700	
		Boeing 737-400	
		ATR42	
		A330-800neo	
		Boeing MD-81	
		Boeing 777F	
		A300-600R	
		Saab 340 B Plus	
		Fairchild Metro/Merlin	
		Beechcraft 1900D	
		A350-1000	
		Embraer 195	
		Embraer 170	

Appendix CDerivation: Mass Growth Factor –Inverse of Payload Fraction

It starts with the Equation (4.12):

$$\frac{\frac{\Delta m_G}{m_{MTO}}}{\frac{\Delta m_L}{m_{MTO}}} = \frac{\frac{\frac{m_{MPL}}{m_{MTO}} + \frac{\Delta m_L}{m_{MTO}}}{1 - \frac{m_{ZP}}{m_{MTO}}} - 1$$

$$\frac{\frac{\Delta m_L}{m_{MTO}}}{\frac{\Delta m_L}{m_{MTO}}}$$

To simplify matters, the following terms are replaced as follows:

$$\frac{\Delta m_L}{m_{MTO}} = y \tag{C.1}$$

$$\frac{m_{MPL}}{m_{MTO}} = x \tag{C.2}$$

Transforming the Equation (4.5) according to m_{MPL}/m_{MTO} leads to:

$$\frac{m_{MPL}}{m_{MTO}} = 1 - \frac{m_{OE}}{m_{MTO}} - \frac{m_F}{m_{MTO}}$$
(C.3)

Thus the Equation (4.6) can be rewritten as:

$$\frac{m_{ZP}}{m_{MTO}} = 1 - \frac{m_{MPL}}{m_{MTO}} \tag{C.4}$$

By inserting the simplification (C.2) it follows:

$$\frac{m_{ZP}}{m_{MTO}} = 1 - x \tag{C.5}$$

The left side of Equation (4.12) corresponds to the mass growth factor. Inserting (C.1), (C.2) and (C.5) into Equation (4.12) leads to a clear presentation:

$$k_{MGW} = \frac{\frac{x+y}{1-(1-x)} - 1}{y}$$
(C.6)

This equation is simplified in the following steps. Dissolving the bracket leads to:

$$k_{MGW} = \frac{\frac{x+y}{x} - 1}{y} \tag{C.7}$$

This matches with:

$$k_{MGW} = \frac{1 + \frac{y}{x} - 1}{y} \tag{C.8}$$

Omission of ones:

$$k_{MGW} = \frac{\frac{y}{x}}{y} \tag{C.9}$$

By shortening it finally follows:

$$k_{MGW} = \frac{1}{x} \tag{C.10}$$

Resetting (C.2) leads to the formula (4.4) for the direct calculation of the mass growth factor:

$$k_{MGW} = \frac{1}{\frac{m_{MPL}}{m_{MTO}}}$$